Preferences and the effectiveness of behavior-change interventions: Evidence from adoption of improved cookstoves in India^{*}

Marc Jeuland

Subhrendu K. Pattanayak Faraz Usmani[†] Jie-Sheng Tan Soo

October 9, 2019

^{*}The authors are grateful to numerous field partners, including field staff at the NGO CHIRAG as well as Vasundhara Bhojvaid, Abhishek Kar, Omkar Patange, Mukul Prakash, Nithya Ramanathan, Rajesh Thadani, Mansi Vora, Ibrahim H. Rehman, Veerabhadran Ramanathan, all of whom were instrumental in helping plan and execute this study. Many graduate student researchers at Duke University also contributed to fieldwork: Jessica Lewis, Nina Brooks, Laura Morrison, Lisa Lipinski. We thank the following in particular for their helpful comments: Fredrik Carlsson, Katie Dickinson, Rema Hanna, Kelsey Jack, Jacob LaRiviere, Dan Phaneuf, V. Kerry Smith, Jennifer Alix-Garcia, and two anonymous reviewers. The study was funded by the United States Agency for International Development under Translating Research into Action, Cooperative Agreement № GHS-A-00-09-00015-00. The study was made possible by the support of the American People through the United States Agency for International Development (USAID). The contents of this publication are the sole responsibility of the authors and do not necessarily reflect the views of USAID or the United States Government. The field experiment described is registered in the AEA Registry for Randomized Controlled Trials (AEARCTR-0003400). Earlier versions of the paper were previously circulated under the title "Preference Heterogeneity and Adoption of Environmental Health Improvements: Evidence from a Cookstove Promotion Experiment," Duke Environmental and Energy Economics Working Paper № EE 14-10.

[†]Jeuland (Email: marc.jeuland@duke.edu; Tel: (+1) (919) 613-4395; Mailing address: Box 90239; Duke University, Durham, NC, 27708, USA) is Associate Professor of Public Policy and Global Health at Duke University and Research Fellow at the RWI–Leibniz Institute for Economic Research, Essen, Germany. Pattanayak is Professor of Public Policy, Global Health, Environment and Economics at Duke University. Tan Soo is Assistant Professor of Public Policy at the Lee Kuan Yew School of Public Policy, National University of Singapore. Usmani is a Postdoctoral Fellow at the Charles H. Dyson School of Applied Economics and Management at Cornell University.

Preferences and the effectiveness of behavior-change interventions: Evidence from adoption of improved cookstoves in India

Abstract

Preference heterogeneity can influence behavior in economically significant ways, thereby influencing the effectiveness of environmental policies or interventions. We test this hypothesis in the context of efficient cooking technology in India. We use stated preference methods to first characterize household tastes for various features of a more efficient cooking technology. We then relate these typically unobserved preferences to households' adoption decisions during an experiment that allowed them to choose between two alternatives with different features. Stated preferences help predict actual adoption: households initially classified as uninterested are less likely to purchase and use any new technology, while relative distaste for pollution is linked to selection of a cleaner technology. Because of this influence on adoption behaviors, preference heterogeneity has important implications for how environmental policies can impact various health and development outcomes.

Keywords: Discrete choice experiment, latent class analysis, field experiment, India *JEL codes*: C93, D12, Q41, Q53

Copyright The Association of Environmental and Resource Economists 2019. Preprint (not copyedited or formatted). Please use DOI when citing or quoting. DOI: https://doi.org/10.1086/706937

1 Introduction

Many policy interventions seek to change private individuals' behaviors for social efficiency or paternalistic reasons. For example, policy-makers often provide incentives or otherwise seek to stimulate adoption of quasi-public goods that improve health, livelihoods, and environmental quality, to overcome private agents' lack of accounting of their external benefits (Foster and Rosenzweig, 2010). Similarly, they may try to help individuals overcome decision-making biases that harm their long-term social welfare (Allcott and Mullainathan, 2010; Ashraf et al., 2006; Madrian and Shea, 2001). Unfortunately, behavior change interventions frequently fall short of desired results, often because targeted beneficiaries are slow to adopt and use new technologies, or to respond to these new policy incentives (Feder et al., 1985).

A potentially significant challenge to implementation of successful behavior change policies, many of which typically prescribe a single mechanism, incentive or technology, arises from deep-rooted (and typically unobserved) demand heterogeneity (Heckman, 2001). Such interventions downplay the potential importance of beneficiary choice. In this paper, we ask a fairly fundamental question about the role and importance of such preferences, namely: Do practical methods exist for measuring preferences that determine beneficiary choice in advance of implementation? To answer this question, we turn to our specific empirical application, which relates household preferences for improved cookstoves (derived from stated preference surveys) to revealed preferences observed when we offer these same households the option of purchasing one or both of two technologies (each with different attributes).

To characterize preferences, we apply generalized multinomial logit methods (Magidson and Vermunt, 2004) to analyze data from a discrete choice experiment (DCE). In this DCE, respondents completed a series of choice tasks in which they considered differences—in price, convenience, smoke emissions, and fuel requirement—between improved cookstoves (ICS) and traditional stoves. In the context of studying demand for cooking technologies, for which well-developed markets do not currently exist, a particular advantage of the DCE is to allow consumers to explicitly consider tradeoffs between potential stove alternatives with varying levels of these attributes (Louviere et al., 2000; McFadden and Train, 2000). We use scale-adjusted latent class analysis (LCA) to look for regularities in the choice patterns of different respondents while accounting for uncertainty in the estimated coefficients across groups (Magidson and Vermunt, 2004). We then consider whether classes of households (categorized through the LCA) with specific preferences are more or less likely to purchase a new stove during a randomized technology promotion campaign. A follow-up survey conducted several months after the intervention sheds additional light on the use of these intervention stoves.

Please use DOI when citing or quoting. DOI: https://doi.org/10.1086/706937 We find that about two thirds (67%) of sample households can be categorized as initially 'uninterested' (we apply this label to the class 3 group) in the positive attributes of ICS. At baseline, these households have lower wealth, are older, and are less aware of the health damages caused by smoke inhalation. The other two classes are distinguished by their relative level of interest in an ICS. Class 1 (11%) is a small, price-sensitive group that is primarily interested in smoke emissions reductions, while class 2(22%) has higher relative demand for the full set of ICS attributes, and particularly for reduced fuel consumption.

Consequently, we observe that class 3 households are much less likely to purchase and use a new stove promoted during a randomized sales campaign (Figure 1). Consistent with predictions from the DCE, class 1 households are relatively more likely to adopt an electric stove that essentially eliminates household emissions. Class 2 households on the other hand are more likely to adopt a biomass-fuel ICS that is slightly more expensive but offers reductions in both fuel needs and emissions. Counterfactual calculations of adoption rates in scenarios in which only one or the other stove would have been offered therefore show substantially lower uptake rates. Importantly, the promotion experiment was designed to address other barriers to adoption of environmental health technologies such as information and liquidity constraints (Bensch et al., 2015), and proved successful in stimulating the acquisition of improved stoves in this population. This perhaps explains why few observable household or community-level covariates other than latent preferences explain stove purchases. Looking beyond stove purchase, we find that differences in adoption rates also translate into differential use, though use rates are mostly similar across groups once we condition on purchase. This latter finding is somewhat limited by statistical power and may stem in part from heterogeneity in treatment effects. These results are robust to a variety of specifications and to dichotomous or continuous probability-based measures of class membership.

We are not the first to argue that latent heterogeneity has important implications for program implementation and policy design (Heckman, 2001; Pinto, 2015). Differences in preferences may arise from variation in beneficiaries subjective expectations about benefits and costs (Manski, 2004; Poe and Bishop, 1999); heterogeneity in risk aversion, time preferences, or trust attitudes (Liu, 2013); differences in tastes and weights related to culture or other factors that beneficiaries place on specific features of a technology (Jeuland et al., 2015a; McFadden and Train, 2000); or from unappreciated complexity in the dynamics that influence technology adoption and the production of benefits (Brown et al., 2017; Jeuland and Pattanavak, 2012). Yet, despite advances in measurement, the role of heterogeneity in influencing outcomes is under-researched (McFadden and Train, 2000; Ravallion, 2012). Indeed, the vast majority of prior work in this vein focuses on attitudes towards risk and time, or trust (Cárdenas et al., 2013; Liu, 2013; Tanaka et al., 2016; Smith and Desvousges, 1986).





Most behavior change interventions meanwhile offer only a single alternative to target populations, for a variety of reasons, including political feasibility, ease of implementation, or paternalism. Such an approach assumes that variation in beneficiaries' attitudes towards interventions and technologies, even if it exists, will not be particularly consequential for determining final outcomes. In this respect, a useful question to ask is: How general are the results that we observe, for one particular environmental health technology such as an ICS? After all, a cleaner cooking stove has complex implications for those adopting it: it alters fuel consumption and requirements, cooking time allocation, respiratory health, household cleanliness, and even the taste of food (Jeuland et al., 2015b). And in general, we would caution that the importance of offering choices likely varies across technologies and settings, and may be particularly important in our specific context.

Reflecting more on this point, the absence of choice may not be a serious concern for interventions for which the pathway from cause to effect is relatively simple (e.g., encouraging vaccination against a specific disease). Still, many social welfare-improving interventions feature complex or indirect pathways to improved well-being. For example, marketing of health-improving water and sanitation technology, such as in-house water chlorination, may reduce convenience, or have aesthetic implications that individuals weight differently (Casabonne and Kenny, 2012; Jenkins and Curtis, 2005; Yang et al., 2006), with important consequences for demand and adoption (Jeuland et al., 2016). Preference heterogeneity has also been observed to affect household demand for sanitation (Vasquez and Alicea-Planas, 2018), vaccines (Larson et al., 2014), insecticide-treated bednets (Bonan et al., 2017), and

Please use DOI when citing or quoting. DOI: https://doi.org/10.1086/706937 farmers' demand for soil fertility management or other ecosystem service enhancements (Lambrecht et al., 2013; Mulatu et al., 2014). Failure to adequately account for perceptions of costs and benefits, and especially those that relate to taste aspects that are usually unobserved, may therefore lead to systematic errors in projections of the demand for, and outcomes obtained from, behavior change efforts (Orgill et al., 2013; Whittington et al., 2012).¹ And while the results that we present in this paper are clearly specific to cleaner cooking technology, this disconnect may have especially important consequences for efficient adoption of multi-dimensional quasi-public goods, because markets that would better offer choices tend to fail or underprovide (Phaneuf, 2013).

Our paper makes two main contributions. First, we add to a thin literature on the private demand for a quasi-public good of importance to economic, environmental, health and energy outcomes—cleaner cooking technologies—by being the first to examine how households respond to a sales offer that allows a choice between two very different technologies. Relatedly, nearly all technology promotion intervention studies in environmental health, and all in the domain of improved cookstoves, focus on the demand for a single pre-selected technology with a specific set of features. Second, we seek to better understand whether and how preference and taste heterogeneity may influence policy outcomes, in terms of both uptake and outcomes among target beneficiaries. The latter contribution speaks to long-standing questions about the reliability of subjective and stated preference data (Bertrand and Mullainathan, 2001; Borghans et al., 2008; Carson et al., 1996). Our findings suggest that stated preference approaches can be used for more than just valuation. That is, stated preferences may serve as a methodological tool for preparing more informative field experiments, and as a policy tool for targetted implementation of programs. These contributions collectively have important implications for policies aiming to promote quasi-public goods and maximize welfare impacts.

The remainder of the paper proceeds as follows: section 2 describes our setting and data; section 3 describes our empirical approach and results concerning preferences; section 4 presents our main findings; and section 5 concludes.

¹Indeed, we demonstrate the extent to which uptake of a policy can differ as a function of such heterogeneity using an illustrative model presented in Appendix A. Specifically, using numerical simulations, we show that in the presence of across- and within-community differences in preferences over policy attributes, the provision of multiple alternatives greatly increases participation by beneficiaries. In our simple set-up, with policies differentiated only along two attribute dimensions, uptake increases by approximately 40% when two differentiated alternatives are made available.

Copyright The Association of Environmental and Resource Economists 2019. Preprint (not copyedited or formatted). Please use DOI when citing or quoting. DOI: https://doi.org/10.1086/706937

2 Research context and data

The use of solid biomass or coal fuels for basic household cooking and heating remains widespread throughout the world, and represents approximately 15% of global energy use (Legros et al., 2009; Smith et al., 2000). Such fuels are often burned in inexpensive yet inefficient stoves, which results in damages to health from respiratory illnesses and other conditions (Bruce et al., 2006; Ezzati and Kammen, 2001; Martin et al., 2011), to local environments and development due to unsustainable and time-intensive harvesting of biomass, and to the global climate as a result of emission of black carbon particles and ozone precursor gases (Bond et al., 2004; Ramanathan and Carmichael, 2008). These negative effects of traditional stoves have prompted great interest in—and a new push towards—development and dissemination of more efficient cook stoves such as gas-, electric-, or cleaner biomass-burning ICS technologies (UN Foundation, 2010).

The target region for this study—in the Northern Indian state of Uttarakhand—is a particularly relevant location for a study of the demand for non-traditional stoves, due to the confluence of several factors: a) growing national and local efforts to disseminate more efficient household energy products; b) increasing awareness and demand for more efficient cooking technologies, stemming from rising fuel scarcity (resulting from population pressure and concern over deforestation) and the health effects of air pollution; and c) location in a region (the Hindu Kush-Himalaya) that is particularly vulnerable to the impacts of climate change. Baseline surveys were conducted in August-October 2012; the promotion intervention occurred from August-October 2013, with follow-up surveys occurring shortly thereafter in November and December 2013.

2.1 Sampling frame

The sampling frame for the study consists of 97 geographically distinct communities (or hamlets) located in 38 Gram Panchayats (GPs) in the Bageshwar and Nainital districts of Uttarakhand (Figure 2). These 38 GPs were drawn from the prior Census of 2,105 GPs using a propensity score matching approach. Specifically, we aimed to create two sample strata, each with 19 GPs, that were mainly differentiated along institutional lines. The first stratum consisted of GPs where the NGO partner for the study had done prior work unrelated to ICS promotion, while the other included villages selected to be observationally similar to the former using nearest neighbor matching based on Census characteristics. The sample therefore may not be representative of rural Uttarakhand; still, one advantage of this design is that it allows us to argue that the results we obtain are not wholly dependent on prior

Copyright The Association of Environmental and Resource Economists 2019. Preprint (not copyedited or formatted). Please use DOI when citing or quoting, DOI: https://doi.org/10.1086/706937 Figure 2: Study design



trust in the implementing institution.

Within each of the 38 GPs, we randomly selected households according to the size of the GP, as further detailed elsewhere (Pattanayak et al., 2019). The sampling strategy ensured that surveys were collected throughout the entire extent of the GP and created variation in the number of hamlets sampled from each GP. The "official" number of distinct hamlets sampled in this way was 106; some of the smallest of these were later combined with nearby hamlets for the purpose of the stove promotion intervention to yield the final set of 97 hamlets.

Efforts were made to interview each sampled household. If a randomly selected household was unavailable during the entire day of baseline fieldwork in a particular hamlet, or if it did not have an eligible respondent (i.e., the primary cook and/or head of the household were unavailable) or refused to participate, neighbors were randomly selected as replacements.² Field supervisors performed household introductions, recorded GPS coordinates and elevation data, and oversaw quality control checks in each village. The final baseline sample consisted of 1,063 households.

 $^{^{2}}$ In total, 118 households were replaced in this way. Thirty-three households refused to participate, while an additional 85 could not be interviewed because an eligible respondent was not home during the field visit.

Copyright The Association of Environmental and Resource Economists 2019. Preprint (not copyedited or formatted). Please use DOI when citing or quoting. DOI: https://doi.org/10.1086/706937

2.2 Surveys and the DCE

The questionnaires used in the baseline surveys included both household and community instruments (completed by a village leader or key informant). Respondents (both the male and female head of household or primary cook were asked to provide specific inputs) answered questions on environmental and stove-related perceptions, household demographic and socio-economic characteristics, stove and fuel use (these questions were supplemented with observations of cooking areas and stoves with clear signs of use, as well as available fuels), and completed the DCE. Follow-up surveys repeated all the aforementioned measurements, except for the DCE. Whenever possible, the primary cook answered questions related to socio-demographics, stove and fuel use, whereas the head of household completed the DCE, socio-economic, and time and risk preference sections. We targeted the head of the household for the latter sections because we judged, based on preparatory focus groups, discussions with our field partners, and prior literature concerning the gendered demand for cooking technology (Krishnapriya and Somanathan, 2019; Miller and Mobarak, 2013), that this individual would ultimately have control over any future decision to adopt an ICS. We also felt that this individual would likely be at least somewhat aware of the preferences of the primary cook as he/she made that purchasing decision. If one of these two was unavailable for the survey (most often the case for male heads of household), the other eligible respondent completed all questionnaire sections. The survey instruments were pre-tested prior to the initiation of fieldwork in approximately 200 households located in nine villages in northern India.

The attributes included in the stove DCE and their levels were selected following a series of eleven focus groups conducted with over 100 respondents in villages similar to sample villages. Attributes retained were fuel requirement, smoke emissions, number of burners, and cost; those eliminated due to lack of clarity or salience to respondents included time savings (fuel savings were deemed easier to understand by respondents), operation and maintenance requirement, fuel loading approach, lifespan of the stove, and the type of fuel allowed. We used SAS software to select efficient combinations of attribute levels for measuring main effects. Important features of the design are summarized in Table 1.

At the start of the stove decision exercise, the ICS technology was described to respondents in detail, and each of the attributes were explained by the enumerator using a specific script accompanied by pictures. After this description, all respondents completed a fourquestion comprehension test. If a respondent answered any questions incorrectly, the relevant description was repeated and the enumerator again verified comprehension before proceeding. Next the respondent was reminded of his/her budget constraint, was told that the ICS options would last three to five years and cost roughly INR 250 per year to maintain, was assured

Please use DOI when citing or quoting. DOI; https://doi.org/10.1086/706937 Table 1: Summary of discrete choice experiment design

Attributes	Levels	Traditional stove level
	500	
Price (INR)	1,000	0
	2,500	
	1	
Fuel requirement	3	3
	4	
	Low	
Smoke emissions	High	Highest
	Highest	
Number of cooling surfaces	1	1
Number of cooking surfaces	2	

Notes. USD $1 \approx \text{INR} 52$ in 2012.

that there were no right and wrong answers, and was reminded that the exercise was purely hypothetical. In each of four choice tasks completed during the survey, respondents were asked to select their preferred option from a set of two ICS alternatives or their existing stove (i.e., neither of the presented ICS). If they selected one of the ICS alternatives, respondents were asked to confirm their willingness to pay the price listed on the card.³ This confirmation was included to decrease the potential for hypothetical bias (Murphy et al., 2005). Following each choice task, debriefing questions were asked to probe the decision-making process and assess the certainty of respondent answers.

2.3 The intervention

Approximately one year after the baseline surveys, the stove promotion intervention was implemented and randomized at the hamlet level; all sample households living in these treatment hamlets were visited by sales teams working for a local NGO; households living in control communities were not (Figure 2). Following careful field piloting of potential

³Specifically, they were asked: "If you had the possibility to purchase this stove at the price stated, would you be willing to make that purchase, if the payment was required at the time of purchase?" Prior to this question, all respondents were reminded to consider their household budget carefully when choosing their preferred option. The specific text in the questionnaire was: "There are no wrong or right answers to these questions. When you make your choice, keep in mind your household budget and your other financial constraints. You should consider carefully whether the benefits of an improved stove would be worth paying for their cost, in terms of stove cost and maintenance requirement. Remember that the improved stoves last three to five years and cost about INR 250 per year to maintain."

Please use DOI when citing or quoting. DOI: https://doi.org/10.1086/706937 promotion techniques (Lewis et al., 2015), trained sales people working in teams of two visited treatment households and conducted intensive promotion activities with them. First, these teams presented treatment households with an information sheet and explanation of the features of the two available stoves (an electric coil stove and a biomass-burning ICS), even as they performed a live tea-making demonstration.⁴ The information sheet (see Appendix B) and demonstration were designed to inform households about the stoves' benefits (reduced smoke, firewood savings, time savings) and costs (price, electricity cost and risk of electric shocks). Then, once the demonstration was complete, the sales people explained the payment and warranty plan. Specifically, all households were given the choice of paying for the stoves upfront or in three equal installments (including a modest financing fee of INR 60 or 80, depending on the stove) that would be collected over a period of four weeks (i.e., in three installments collected two weeks apart). The stoves were protected by a one-year warranty, that households could access by contacting the NGO if they had problems. Roughly two-thirds of purchasing households opted to pay for their chosen stove in installments.

In addition, households were told that they would receive a randomized rebate at the time of the final payment if they were found to be using the stoves (as observed during unannounced visits). Those paying for stoves upfront were also eligible for the rebate and thus were also revisited at that time. Prior to households indicating whether they would purchase a stove, this randomized rebate was revealed by drawing a chit out of a bag. The bag contained equal numbers of chits corresponding to the three potential rebate levels, low—INR 25 (roughly 2.5%), medium—INR 200 (roughly 20%), and high—equivalent to a full installment (a 33% discount). The electric stoves were sold to households for INR 900 (or INR 960 for those paying in installments); biomass stoves were INR 1,300 (or INR 1,380 with installments). These prices corresponded to the stove-specific prices paid to suppliers. As such, the amount of the high rebate (INR 320 or 460) varied somewhat based on the stove that was chosen by a household. Due to concerns over the endogeneity of the high rebate amount, we replace this varying amount with INR 320 in our analyses (the rebate for the electric stove); none of our results are sensitive to this approach.⁵ Finally, because of this design and the two follow-up visits to intervention communities that it entailed, households that declined a stove during the first visit were allowed to purchase one during follow-up visits so long as they caught up with the installment payments they had missed.

We opted for this intervention design based on both small-scale piloting experiences in

⁴We offered two types of biomass ICS in initial piloting activities, but it quickly became apparent that demand for one of these technologies was low. After observing great interest in a similarly-priced electric stove in later pilots, we decided to offer it alongside the more affordable of the biomass-burning stoves.

⁵The sensitivity of purchases to the rebate level that we estimate may thus be somewhat overestimated, particularly for the biomass stove, but the difference in values is qualitatively small.

Please use DOI when citing or quoting. DOI: https://doi.org/10.1086/706937 eight villages and on initial analyses of the DCE responses. The latter revealed substantial heterogeneity in overall demand, as well as relative weighting that households gave to smoke reductions, which were greatest with the electric stove, versus fuel savings, for which the biomass ICS was better given the cost of electricity (Bhojvaid et al., 2014; Jeuland et al., 2015a). This evidence on heterogeneous preferences gave us reason to believe that artificially constraining the choice set by offering one stove prescriptively in all intervention communities might depress demand.

It might seem strange that a household would opt for a seemingly more rudimentary biomass ICS when offered a modern electric stove that was itself slightly cheaper, but there are a number of reasons why the latter may not strictly dominate in this context. First, the relative cost of electricity—or at least its perceived relative cost—may be considerably higher for some, particularly those households consuming in higher blocks of the increasing block tariff for electricity pricing, those with a lower shadow value of fuel collection time, or those for whom time costs are less salient than pecuniary costs. The fact sheet emphasized that households should expect that their bill would increase by at least INR 260 per month if they used the stove for one hour each day. Pushing in the opposite direction, some households did not have working meters or had irregular electricity connections; these consumers would know that they essentially face zero marginal cost for additional electricity consumption. Second, electricity reliability varies across time and space in Uttarakhand, and different households may have varying tolerance for outages. Third, the perceived durability of the stoves might be a source of heterogeneity; in this respect, we did not have data on durability prior to the intervention but did find the electric stove to be more prone to breakage (e.g., regulator knobs falling off, wires becoming loose). This was also manifest in a risk of electric shocks, which were mentioned in the fact sheet. Fourth, the portability of the electric stove was limited due to the required proximity of a power outlet. And finally, qualitative work showed that households believed that biomass stoves provided better-tasting food, especially for bread (Bhojvaid et al., 2014). This list of issues is also reflective of observations made about divergent preferences for different ICS in other contexts (Hollada et al., 2017).

On the basis of power calculations and our estimation of the differential treatment effects expected from the alternative rebate levels, 70 of the baseline hamlets (corresponding to 762 of the 1,063 baseline households) were randomly assigned to the treatment group.⁶ The remaining 27 hamlets were control hamlets that only received survey visits, at baseline and follow-up (Figure 2).

⁶Two of these 762 treatment households could not be re-located during the stove promotion campaign, and therefore do not appear in any results

Copyright The Association of Environmental and Resource Economists 2019. Preprint (not copyedited or formatted). Please use DOI when citing or quoting. DOI: https://doi.org/10.1086/706937

2.4 Sample balance and descriptive statistics

The analyses in this paper use data collected at three points in time: at baseline (all study households), at the time of the intervention (treatment households only), and three months after the intervention (all study households who could be relocated). The intervention data include only basic information on whether a household purchased a stove, which one it chose, the randomly-assigned rebate level, and the specific payment made during each visit from the sales team. In the post-intervention survey with all households, as discussed in Section 2.2, we again collected detailed information on outcomes such as households' stove ownership, and use of stoves and fuels. Thus, we analyze the DCE data that pertain to the entire sample of households that includes treatment and control communities to characterize preference patterns, then analyze purchase of stoves for the entire sample to account for any adoption of the biomass and electric ICS that might have occurred independently of our intervention (we find none), but only differentiate use results by class for households in the treatment communities, given the lack of adoption of intervention stoves among controls.

Descriptive statistics from the baseline sample of 1,063 households (overall and disaggregated by treatment group) are summarized in Table 2. In 73% of surveys, the main respondent for all questions was a woman (primary cook and/or female head of household). Interviews with the remaining 27% generally included both a male head of household and the primary cook, according to the assignments described above. The average household size at the time of the survey was 4.8 people. Sample households are generally rural, poor, and primarily agricultural. Over half of the survey population reported being below the poverty line, and access to credit was low (with just 15% of households availing of credit in the prior year). Almost all had electricity, but only 25% reported having continuous power, and the average hours of supply per day were about 17.

At the time of the baseline interviews, nearly all households had a traditional mud stove (40%) or three-stone stove (49%). Other commonly-found stoves were LPG (28%), or a traditional metal *sagarh* stove (21%). The average number of stoves owned by each household was 1.4. Nearly all (93-98%) households owning LPG and traditional stoves reported using these in the week prior to the survey, and almost all LPG-owning households used it alongside a biomass stove (only 7% of these did not also use their traditional stoves daily). Households reported total stove use time to be 5.7 hours per day, and solid fuel collection time was also substantial at 1.8 hours per day. Respondents identified that the three best aspects of traditional stoves were: the taste of the food (90%), the cost of the stove (55%), and the ability to cook all foods (7%). The four worst features identified were the smoke that is produced (63%), the cleaning requirements (45%), and the amount of fuel required and the

Copyright The Association of Environmental and Resource Economists 2019. Preprint (not copyedited or formatted). Please use DOI when citing or quoting. DOI: https://doi.org/10.1086/706937 heat given off by the stove (22% each).

The most commonly used fuels by households, many of whom regularly used multiple types, were firewood (97%), LPG (28%) and kerosene (8%), the latter primarily as a lighter fluid. The main respondent in each household was asked whether he/she had heard or knew about each of three negative impacts of traditional stoves and biomass fuels, on health, on local forests, and on air quality and/or climate. Awareness of the negative health effects was highest (62%), followed by local environment and forests (58%), with only 39% recognizing air pollution and/or climate change. Women or primary cooks reported greater awareness of these impacts, but knowledge of ways to mitigate them was limited. Only 25% of respondents said they had heard of stoves that produce less smoke than others at the time of the interview, and only 31% believed that some fuels produce less smoke than others when burned.

The treatment and control households were well balanced across a number of key variables measured at baseline, as shown in column (4) of Table 2. Normalized differences are modest, and only three out of more than 50 variables—female head of household, reported price of LPG, and a belief that ICS can deliver environmental benefits—are significantly different at the 10% level when the variable is regressed on treatment status.⁷

3 Empirical analysis of preferences

3.1 Modeling

We begin our analysis by sorting respondents into preference groups using the baseline stated preference data collected prior to our stove promotion experiment. The framework for analyzing this DCE data is based in random utility theory. We model repeated household choices across the combinations of stove alternatives that vary according to well-defined levels of four specific attributes: price, fuel requirement, smoke emissions, and the number of cooking surfaces. The random utility model we apply assumes that the indirect utility associated with a particular alternative can be written as a function of its attributes and household characteristics:

$$U_{jt}^{i} = V^{i} \left(p_{jt}, \beta_{0}^{i}, X_{jt}, \beta^{i}, Z^{i} \right) + \epsilon_{jt}^{i} \tag{1}$$

 $^{^{7}}$ In the appendix, we also show that the rebate level—randomized across all treated households—is generally uncorrelated with baseline household characteristics (Appendix Table C1). No normalized differences across these groups exceed 0.2 and 8 out of 87 analyzed coefficients are significant at the 10% level, similar to the proportion that would be expected due to chance.

Variable	(1) Mean	(2) Mean	(3) Mean	(4) Normalized difference
	(Overall)	(Control)	(Treatment)	(Treatment – Control)
Village has naved road	0.30	0.26	0.31	0.106
Distance to doctor (km)	9.28	9.16	9.33	0.021
Bank facility in village	0.32	0.33	0.32	-0.011
Presence of implementing NGO	0.50	0.43	0.53	0.190
Household size	4.83	4.98	4.77	-0.101
Number of children under five	0.47	0.54	0.44	-0.118
Age: Head of household (years)	54.0	53.9	54.0	0.011
Education of household head (years)	6.21	6.17	6.23	0.014
Education of primary cook (years)	4.76	4.74	4.76	0.006
Female head of household	0.27	0.32	0.25	-0.150*
Main respondent: Head of household	0.53	0.58	0.51	-0.157
Main respondent: Primary cook	0.77	0.75	0.78	0.060
Hindu	1.0	1.0	1.0	0.00
General caste	0.72	0.71	0.73	0.042
Scheduled caste or tribe	0.25	0.24	0.26	0.047
Household members cold/cough in past two weeks	0.073	0.063	0.077	0.079
Below poverty line household	0.57	0.60	0.56	-0.089
Perception of relative wealth (1: Low to 6: High)	2.13	2.12	2.13	0.010
Household took loan in past year	0.15	0.12	0.16	0.121
Household can save money	0.25	0.24	0.26	0.028
Constant electricity supply	0.25	0.27	0.24	-0.067
Intermittent electricity supply	0.72	0.70	0.72	0.042
Hours of electricity per day	17.3	17.9	17.0	-0.116
Log of total expenditure (INR/month)	8.41	8.37	8.42	0.065
Number of cell phones	1.30	1.29	1.30	0.010
Total rooms in house	4.63	4.43	4.70	0.113
Household has private toilet	0.85	0.89	0.84	-0.123
Household owns/leases agricultural land	0.97	0.94	0.98	0.253
Respondent is most patient	0.48	0.44	0.50	0.125
Respondent is most risk-taking	0.45	0.41	0.45	0.042
Respondent aware of clean stoves	0.20	0.24	0.20	0.050
Respondent holieves alean stoves/fuels can reduce possive impacts	0.31	0.29	0.32	0.051
Respondent believes Cean stoves/rules can reduce negative impacts	0.30	0.27	0.11	0.044
Respondent believes ICS deliver benefits for local forests /environment	0.11	0.10	0.11	0.044
Respondent believes ICS deliver air quality/climate henefits	0.20	0.23	0.20	0.052
Respondent believes smoke is unsafe	0.50	0.47	0.51	0.076
Household owns traditional stove	0.97	0.96	0.98	0.138
Household owns improved stove	0.31	0.29	0.31	0.047
Traditional stove use (minutes/day)	289	306	282	-0.165
Improved stove use (minutes/day)	44.9	40.0	46.8	0.073
Use firewood	0.97	0.96	0.98	0.098
Use kerosene	0.081	0.066	0.087	0.077
Use LPG	0.28	0.26	0.29	0.062
Use electricity	0.01	0.003	0.007	0.041
Use biogas	0.01	0.010	0.009	-0.012
Amount of firewood used (kg/day)	7.68	8.09	7.52	-0.104
Time spent collecting solid fuels (hours/day)	1.84	1.62	1.93	0.189
Price LPG cylinder (INR 1,000)	451	443	455	0.191*
Price of fuelwood $(INR/100 \text{ kg})$	626	603	635	0.049
Total fuel expenditure (INR/month)	247	247	247	0.001
Preference class 1 - Low demand	0.11	0.12	0.11	-0.043
Preference class 2 - High demand	0.22	0.25	0.21	-0.088
Preference class 3 - Uninterested	0.67	0.63	0.68	0.106
Sample size: Households	1063	771	292	
Sample size: Hamlets	97	70	27	

Table 2: Baseline descriptive statistics and balance tests, by treatment group

Notes. "Traditional" stoves include: mitti ka chulha (mud stove), anjeti, three-stone fire, and sagarh (coal stove). At the time of the baseline survey in 2012, USD 1 \approx INR 52. Balance was also assessed by regressing each variable in the left-hand column on treatment status using OLS, clustering standard errors at the hamlet level. Significance of the coefficient for treatment status from these regressions is indicated in column (4) as follows: *** p < 0.01; ** p < 0.05; * p < 0.1.

Please use DOI when citing or quoting DOI: https://doi.org/10.1086/706937 where U_{jt}^i represents household *i*'s utility from cooking alternative *j* in a choice set, and *t* indexes the number of choice tasks completed (four per household); $V^i(\cdot)$ is the non-stochastic portion of the utility function for household *i*; p_{jt} is the price of cooking alternative *j* in task *t*; β_0^i is a parameter that represents household *i*'s marginal utility of money; X_{jt} is a vector of non-price attribute levels for cooking alternative *j* in task *t*; β^i is a vector of parameters that represent the marginal utility for household *i* associated with the different non-price attributes of the alternatives; Z^i is a vector of characteristics for household *i*; and ϵ_{jt}^i is a stochastic disturbance term.

Assuming that households maximize utility within a given choice task, they will select alternative j from among the set of K alternatives presented to them if and only if that alternative provides a higher overall level of utility than all others, that is, if $U_{jt}^i > U_{kt}^i$ for all $j \in K$, where $j \neq k$, such that $V_{jt}^i - V_{kt}^i > \epsilon_{kt}^i - \epsilon_{jt}^i$. With a linear specification of utility $U_{jt}^i = \beta^i X_{jt} + \beta_0^i p_{jt} + \epsilon_{jt}^i$ and a Type 1 extreme-value error distribution for the disturbance term, the probability that alternative j will be selected from choice set t corresponds to the standard conditional logit (McFadden, 1981), which is estimated using maximum likelihood.

We relax the restrictive assumption of the conditional logit that requires a single set of fixed β coefficients, and estimate two types of mixed (or random parameters) logit models.⁸ The first is the mixed logit, which allows for unobserved heterogeneity in tastes across individuals, as specified through inclusion of respondent-specific stochastic components η^i for each of the estimated coefficients β in the model. In the mixed logit model, the choice probability function is described by the integral of the product of conditional individual probabilities over all choice occasions t, where the marginal utilities for different attributes are measured by β^{i*} :

$$\mathbb{P}\left[C^{i} = \left(C_{j1}^{i}, \dots, C_{jT}^{i}\right)\right] = \int \prod_{t} \frac{\exp\left(\beta^{i*}X_{jt} + \beta_{0}^{i*}p_{jt}\right)}{\sum_{k=0}^{K} \exp\left(\beta^{i*}X_{kt} + \beta_{0}^{i*}p_{kt}\right)} f\left(\eta \mid \Omega\right) \,\mathrm{d}\eta, \tag{2}$$

where $\beta^* = (\beta + \eta^i)$ and $f(\eta | \Omega)$ denotes the density of the individual disturbance term η^i given the fixed parameters Ω of the distribution. The stochastic portion of utility flexibly accommodates correlations across alternatives and choice tasks. The coefficients β^* are estimated using simulated maximum likelihood (Revelt and Train, 1998). The ratios of coefficients derived from the model then yield the monetized marginal utility to household *i* for an additional unit of a given attribute. This mixed logit model is useful for understanding the overall heterogeneity in preferences for different attributes within our sample.

⁸There are several problems with the conditional logit, including violation of the independence of irrelevant alternatives (IIA) assumption, the inability to account for correlation across a respondent's choices, and the lack of consideration of differences in individual tastes other than those related to the specified attributes of alternatives.

Please use DOI when citing or quoting. DOI: https://doi.org/10.1086/706937 The second model we estimate is the scale-adjusted latent class logit, a discrete version of the mixed logit that allows us to examine the extent to which respondents can be grouped into distinct sub-populations with similar preferences. This model allows for two sources of heterogeneity (Magidson and Vermunt, 2004; Thiene et al., 2014). First, households with similar choice patterns and weighting of attributes are categorized into preference classes. Second, households within each preference class are further classified into scale groups, which indicate the level of uncertainty of their classification within a preference class.⁹

In these models, maximum likelihood methods are used to identify C preference class types and S scale groups (where C and S are integers selected by the modeler).¹⁰ The probability of observing respondent i selecting alternatives j over T choice tasks is written as a product of the probability of the respondent belonging to preference class c and scale group s:

$$\mathbb{P}\left[C^{i} = \left(C_{j1}^{i}, \dots, C_{jT}^{i}\right)\right] =$$

$$\prod_{t=1}^{T} \left[\sum_{s=1}^{S} \sum_{c=1}^{C} \left(\frac{\exp\left(\alpha_{c} Z_{i}\right)}{\sum_{c=1}^{C} \exp\left(\alpha_{c} Z_{i}\right)}\right) \left(\frac{\exp\left(\gamma_{s} Z_{i}\right)}{\sum_{s=1}^{S} \exp\left(\gamma_{s} Z_{i}\right)}\right) \left(\frac{\exp\left(\lambda_{s} \beta_{c} X_{jt} + \lambda_{s} \beta_{0,c} p j t\right)}{\sum_{k=0}^{K} \exp\left(\lambda_{s} \beta_{c} X_{kt} + \lambda_{s} \beta_{0,c} p_{kt}\right)}\right)\right].$$
(3)

The first two terms on the right hand side of equation (3) are the unconditional probabilities of membership in preference class c and scale group s, respectively, while the third term corresponds to the probability of choosing a sequence of alternatives based on their attributes (e.g., β_c is a vector of coefficients for the attributes of the alternatives and λ_s is the scale parameter for scale group s). In most latent class applications, the vector Z_i in the first and second term includes characteristics such as age, education level, and income. Here, we omit these characteristics and replace Z_i with a vector of class and scale-specific constants α_c and γ_s because our goal is to use only information revealed by the choices households make in the DCE to predict adoption of cleaner stoves.¹¹ This modeling approach thus requires very few assumptions about the structure of stated preferences. In the last term, the taste parameters (β) are subscripted with the class indicator c, meaning that every respondent in class c has identical tastes for the attributes of the choice alternatives in the DCE.

Rather than assuming a specific number of classes and scales, we rely on the Bayesian

⁹The role of the scale parameter in estimating logit models was first highlighted by Swait and Louviere (1993). Magidson and Vermunt (2007) later extended this work to the LCA framework, showing that different scale parameters can confound interpretation of the part-wise utilities across latent classes.

 $^{^{10}}$ See Appendix D for a more detailed explanation of latent class logit models, and a technical description of the scale-adjusted model.

¹¹If socioeconomic characteristics were also included in the estimation of class membership probabilities, these predicted probabilities would then partially reflect these observable factors. This would conflate our investigation of the role of latent preference heterogeneity in technology adoption.

Please use DOI when citing or quoting. DOI: https://doi.org/10.1086/706937 Information Criterion (BIC) to select the best-fitting model with up to six different classes and three scales (Roeder et al., 1999). We then assign a household to the particular class for which its probability of membership is greatest, and study the correlates of class membership using a multinomial logit model.

3.2Simple mixed logit analysis without preference classes

We first analyze the heterogeneity in preferences for ICS attributes. We estimate two mixed logit models (Table 3), which are differentiated by the assumed distribution of the random coefficient for price: either fixed (Columns 1 and 2) or log-normal (Columns 3 and 4).¹² The coefficients for the attributes all have the expected signs: alternatives with higher prices, emissions and fuel requirements were less likely to be selected by respondents, whereas those with a greater number of cooking surfaces or of traditional type were more likely to be selected (all other attributes being equal). The standard deviations of the random parameters for the traditional stove type and for price are significant (Columns 2 and 4). In terms of magnitude of effects, comparison of the part-wise utilities for a single unit change in the levels of the various attributes suggests that the value of a one-unit (33%) reduction in smoke emissions and additional cooking surface are similar on average, followed by a one-unit (33%) decrease in fuel requirement. The large coefficient on the traditional stove type indicates an average preference for traditional stoves that outweights the value of a one-unit reduction in smoke emissions plus fuel consumption several times over; this implies that respondents on average would need to see large reductions in these levels to consider adopting an ICS.

3.3 The preference classes and their determinants

Given the heterogeneity in responses identified by the random parameters model, we next use LCA to look for consistent patterns in the choices made by different groups of respondents, and to test the extent to which they are associated with observable household and respondent characteristics. In the 3-class and 2-scale model with the best fit according to the BIC, classes 1 ($\sim 12\%$ of respondents) and 2 ($\sim 23\%$) react negatively to increased fuel usage and smoke emissions, and react positively to increased cooking capacity (Table 4). Given that cleaner cooking technologies are supposed to reduce emissions and fuel requirements, we might expect these two classes to be more likely to adopt them.¹³ In contrast, class 3 ($\sim 65\%$) is an 'uninterested' group; none of the stove attribute coefficients for this group are significant,

 $^{^{12}}$ By restricting the distribution of the price coefficient in these ways, we ensure that price will be negatively related to the adoption decision.

 $^{^{13}}$ Some alternative stoves also have multiple cooking surfaces, though the ones we promoted during this study did not.

Variables	Fixed 1	orice	Lognorm	al price
variables	(1) Mean	(2) SD	(3) Mean	(4) SD
Price $(INR)^{\dagger}$	-0.239***		-1.03***	2.53***
	(0.026)		(0.196)	(0.264)
Fuel requirement	-0.143***	-0.043	-0.158***	0.147^{***}
	(0.036)	(0.209)	(0.039)	(0.148)
Smoke emissions	-0.350***	-0.046	-0.368***	0.071
	(0.087)	(0.269)	(0.091)	(0.172)
Number of pots	0.358^{***}	0.099	0.389^{***}	0.260
	(0.084)	(0.455)	(0.090)	(0.282)
Traditional stove [‡]	2.76^{***}	4.52^{***}	1.32^{***}	4.19^{***}
	(0.276)	(0.295)	(0.322)	(0.326)
Part-wise utility as	sociated wi	th one-ui	nit decrease	$(USD)^{\$}$
Fuel requirement	\$5.8		\$4.3	
Smoke emissions	\$14.1		\$9.9	
Number of pots	-\$14.4		-\$10.5	
Observed choices	9,16	2	9,16	52
Likelihood ratio (χ^2)	1,278	8.0	1,330	5.6

Please use DOI when citing or quoting. DOI: https://doi.org/10.1086/706937 Table 3: Mixed logit analysis of DCE choices

Notes. *** p < 0.01, ** p < 0.05, * p < 0.1; Standard errors in parentheses. Model excludes respondents who answered any one of four comprehension questions incorrectly prior to the first choice task. [†]Price is in INR divided by 500 (USD 1 \approx INR 52 in 2012), and -500 in the logged version. [‡]Traditional stove type = 1 if it was the traditional stove, 0 if improved. [§]One unit in the DCE represents 33% of traditional stove smoke emissions and fuel consumption, and a single cooking surface.

except for its clear preference for traditional stoves. Comparing classes 1 and 2, class 1 is considerably more price sensitive, but is relatively more responsive to smoke emissions reductions (the implied part-wise utility associated with a one-unit smoke emissions reduction is still lower than that for class 2, however), whereas class 2 places greater relative weight on fuel reduction and convenience. All else being equal, both of these classes also prefer improved biomass stoves to traditional stoves, as shown by the alternative-specific constant (ASC). Based on these households' choice patterns, we can consider class 2 to be a high demand group, class 1 to be low demand, and class 3 to be much less interested by the ICS.

Since class 3 constitutes about two thirds of the sample, it is important to note the possibility that such respondents may not have understood or paid attention to the DCE. Yet the pattern of this group's responses suggests that they were not answering questions haphazardly; rather, they simply tended to prefer the traditional alternative, no matter the

Variables	(1) Class 1 Low demand	(2) Class 2 High demand	(3) Class 3 Uninterested
Price $(INR)^{\dagger}$	-1.95***	-0.13***	-0.50
	(0.50)	(0.048)	(0.36)
Fuel requirement	-0.39**	-0.26***	0.060
	(0.17)	(0.071)	(0.24)
Smoke emissions	-0.93**	-0.46**	0.22
	(0.39)	(0.18)	(0.66)
Number of pots	1.01^{**}	0.67^{***}	-0.65
	(0.41)	(0.16)	(0.73)
$ASC - Traditional stove^{\ddagger}$	-1.60**	-2.44**	3.49**
	(0.75)	(0.96)	(1.46)
Fraction of households in class	0.11	0.22	0.67
(based on predicted probability from LCA)			
Fraction of class in scale group 1^{\S}	0.91	0.85	0.82
(based on predicted probability from LCA)			
Observations	9,162	9,162	9,162

Table 4: Latent class analysis of DCE data for improved biomass cookstove

Notes. *** p < 0.01, ** p < 0.05, * p < 0.1; standard errors in parentheses. [†]Price is in INR divided by 500 (USD 1 \approx INR 52 in 2012). [‡]This is the alternative-specific constant: Traditional stove type = 1 if it was the traditional stove, 0 if improved. [§]The two scale parameters are 1 (class 1) and 0.16 (class 2).

attributes of the ICS alternatives.¹⁴ This is further shown by the large relative size of the price coefficient, and the large positive coefficient on the ASC for the traditional stove.

We also note that the estimation identifies two scale classes that are defined by their respective scale parameters of 1 (following normalization) and 0.16. These parameters can be interpreted as indicating greater uncertainty in the responses for those in the second scale class (scale parameter=0.16). The overall proportion of respondents in scale class 1—as obtained from the predicted probability derived from the posterior distribution estimated in the model—is 83.7%, with the remaining 16.3% belonging to scale class 2.

To further investigate the characteristics of these classes, we assigned each respondent household to the class for which it had the highest predicted probability of membership, as obtained from the LCA. We then regressed this predicted class on a variety of demographic and socio-economic variables using a multinomial logit model, where all reported coefficients are relative to omitted class 3 (Table 5). We observe that, in comparison to class 3, classes 1 and 2 are generally wealthier, have better educated household heads (especially class 2), and better able to save, and for class 1, have younger heads of household. This is consistent

¹⁴We determined that many of these households were serial non-responders, in the sense that they always chose the traditional stove (see Appendix Table C2).

Please use DOI when citing or quoting. DOI: https://doi.org/10.1086/706937 with earlier research that finds similar factors to be positively associated with adoption of clean stoves (Lewis and Pattanayak, 2012). Both classes are also more aware of clean stoves, especially class 1; the latter furthermore report high firewood prices and use traditional stoves somewhat less (though the latter difference is not statistically distinguishable from zero). Comparing between classes 1 and 2, we observe that class 2 is wealthier, which may also explain the lower price sensitivity of such households and their higher willingness to pay for all three ICS attributes. Class 2 respondents are also the most patient (as judged by responses to hypothetical time preference questions). If future health benefits of using improved stoves are most meaningful to these respondents, this may further contribute to their lower price sensitivity.

Empirical analysis of stove adoption and use 4

4.1Modeling

From the stove promotion campaign and follow-up surveys conducted several months later, we observe households' purchase and use decisions. We regress these outcomes on latent class membership as identified based on the responses in the DCE. The most general model we estimate can be written as:

$$Y_{ij} = \beta_0 + \beta_1 \cdot T_{ij} + \beta_k \cdot C_{kij} + \beta_r \cdot r_{ij} + \beta_{kr} \cdot C_{kij} \cdot r_{ij} + \beta_c \cdot X_{mij} + \mu_j + \epsilon_{ij}.$$
(4)

In this model, Y_{ij} is an indicator variable representing purchase or use of an intervention stove by household i in community j. More specifically, we analyze purchase during the initial sales visit and at the end of the campaign, and use observed at the time of the follow-up survey visit. The variable T_{ij} is an indicator variable that equals 1 if the household i is in the treatment group and 0 otherwise, and C_{kij} is equal to 1 if the household i has preferences of type k and 0 otherwise (as revealed by the LCA)¹⁵; r_{ij} represents a rebate amount randomized at the household-level in the communities exposed to the stove offer; X_{mij} is a vector of m household and community variables that influence the purchasing decision (the same set of variables used in the multinomial logit described above); μ_i is an error term clustered at the

 $^{^{15}}$ To test for sensitivity of our results to the definitions of class membership, we also estimated the same models using the continuous probabilities of class membership generated by the latent class logit procedure. We choose to present the dichotomous class indicators only because the interpretation of coefficients is more straightforward. The results with continuous probabilities are qualitatively identical to those presented in the paper, and are available in Appendix E. We also estimated models that controlled for the scale grouping; this covariate was never statistically significant, and likewise did not change any of the results (results available upon request).

Please use DOI when citing or quoting. DOI: https://doi.org/10.1086/706937 Table 5: Correlates of latent class membership

Variables	(1) Class 1	(2) Class 2
Presence of implementing NGO	0.32	0.083
	(0.26)	(0.24)
Household size	-0.032	0.022
	(0.073)	(0.055)
Household has child under five	-0.017	0.27
	(0.30)	(0.21)
Age of household head	-0.030***	0.003
	(0.010)	(0.006)
Education of household head	0.029	0.058**
	(0.028)	(0.023)
Education of primary cook	-0.006	0.003
	(0.026)	(0.019)
Female household head	0.39	0.19
	(0.30)	(0.26)
Respondent is primary cook	-0.22	-0.26
	(0.28)	(0.21)
General caste	-0.44	-0.27
	(0.30)	(0.25)
Proportion of household sick with cough/cold	0.068	-0.42
in past two weeks	(0.53)	(0.69)
Perception of relative wealth	0.092	0.46***
	(0.17)	(0.11)
Household took loan in past year	0.079	0.26
Household tooli louir in past your	(0.29)	(0.21)
Household can save money	0.76***	0 74***
Household can save money	(0.25)	(0.22)
Household has private toilet	-0.024	-0.61*
	(0.30)	(0.24)
Respondent is most patient [†]	-0.36	0.59***
respondent is most patient	(0.31)	(0.23)
Respondent is most risk-seeking [†]	0.40	-0.12
respondent is most risk-seeking	(0.37)	(0.21)
Respondent aware of clean stoves	1 0***	0.34
respondent aware of clean stoves	(0.38)	(0.42)
Respondent believes clean stoves/fuels can	0.20	0.52
reduce nogative impacts	(0.25)	(0.38)
Respondent believes smoke is unsafe	-0.10	-0.33*
respondent beneves smoke is unsale	(0.24)	(0.17)
Household owns improved stove	0.24)	(0.17) 0.27
Household owns improved stove	(0.31)	(0.21)
Traditional stove use (hours/day)	0.075	0.030
fractional stove use (nours/day)	(0.065)	(0.030)
Log of time grant collecting solid fuels	(0.003)	0.057
Log of time spent conecting solid fuels	(0.014)	(0.057)
Respondent reports high price of freewood	0.62**	0.80***
Respondent reports high price of mewood	(0.25)	(0.20)
Constant	0.20	0.20
Oustallt	-0.19	(0.57)
	(0.00)	(0.07)
Observations	987	987
Pseudo- R^2	0.1	25

Notes. Multinomial logit specification, class 3 is the omitted class. *** p < 0.01, ** p < 0.05, *p < 0.1; standard errors in parentheses. †Most patient and most risk-seeking as determined by responses to three hypothetical time and risk preference questions.

Please use DOI when citing or quoting. DOI: https://doi.org/10.1086/706937 community level; and ϵ_{ij} is the usual individual idiosyncratic error term. The coefficients β are estimated using OLS, and allow us to consider the effects of preferences and price incentives (β_k and β_r , respectively) on outcomes.

In the purchase models (which shed light on the general patterns shown in Figure 1), we first group the improved biomass and electric stoves into one general category and analyze adoption of any promotion stove, using a linear probability model. We then apply a multinomial logit model that treats the three options as a categorical outcome for each household (no stove, electric, or improved biomass stove). We consider more parsimonious specifications for equation 3 as well as the complete model. In the most basic model, we include only the binary predicted class variables to explain purchase. We then add the randomized rebate amounts, which have a very large effect on purchasing rates (Figure 3), followed by a comprehensive set of community and socioeconomic characteristics, including those used in Table 6 to predict class membership. To better interpret whether class membership adds meaningful explanatory power to the model, we compare the fit (using an F-statistic) for the specification that includes class indicators and the full set of controls to that with the full set of covariates alone.¹⁶ In addition to the regression analyses, we derive choice probabilities from the DCE for each stove-preference class combination using their relative attribute levels, in order to make counterfactual predictions of the adoption levels that would have been observed in the absence of stove choice.

Standard errors in all analyses are clustered at the community or hamlet level as this is the administrative level at which the stove promotion campaign was assigned.

4.2Analysis of stove adoption and use

Below we consider two main questions on the relationship between stated preferences and adoption. The primary variables of interest in much of the discussion that follows are the binary variables for membership in each of the three latent classes. In the ensuing discussion, we report results from all estimated models but our preferred specification is the full model. This is because inclusion of other controls allows us to better isolate the effects of unobserved preferences from the more commonly observed drivers of clean stove ownership documented in the literature (Lewis and Pattanayak, 2012).

¹⁶We also explore whether information on these more easily observable variables might facilitate systematic targeting of households more likely to adopt an ICS. To do so, we use a two-stage least squares model that generates predictions of preference class as a function of the observed covariates among the treated, and relate those predicted classes to actual purchases.





Question 1: Is preference class related to adoption of a new stove?

This question arises from the observation that the three preference classes responded very differently to the stove attributes in the DCE exercise. We consider two separate purchase variables: at first contact with the sales team, and then over the course of the entire sales campaign, with a revised purchasing outcome that includes lagging purchasers.

The results show that compared to class 1, class 3 households were about 18-20 percentage points less likely on average (or a roughly 33% lower purchasing rate) to purchase a new stove during the first sales visit (Table 6, columns 1-3). Counter to our initial expectations given their different price sensitivities in the DCE, we do not detect statistically meaningful differences between class 1 and class 2 households with respect to this purchase decision, though the coefficient for class 1 is substantially larger.¹⁷ The rebate amount itself had a strong positive effect on stove purchase: an increase from the low rebate of INR 25 (about 2% of stove cost) to the high rebate (33% of stove cost) increased purchase from 28% to about 72% (Figure 3). No other controls are significantly related to purchase, including those for relative firewood cost and average hours of electricity supply. The *F*-statistic comparing the model fit for the model with *a*) classes plus controls versus *b*) controls only is highly

 $^{^{17}}p$ -values for significance on differences in these coefficients range between 0.24 and 0.44. The lack of difference may be because class 1 purchased relatively more electric stoves as rebate level increased, while class 2 purchased relatively more biomass ICS. We also interacted the rebate level with the class membership to test for differential sensitivity to price; those results suggest that classes 1 and 2 were more responsive to price discounts than class 3, although differences based on a Wald test are not significant across specifications (see Appendix Table C3). One additional rupee of rebate increases the probability of class 1 and 2 households purchasing stoves by 0.16-0.20% on average, compared with a marginal impact of 0.14% for class 3.

Please use DOI when citing or quoting. DOI: https://doi.org/10.1086/706937 significant (F = 9.54; p < 0.001), while that comparing the fit with a) classes plus controls versus b) classes only is not (F = 1.04; p = 0.41). These comparisons of model fit and the stability of the coefficients on class membership suggest that the technological preferences expressed in the DCE contain information that is not reflected in more readily observable characteristics. And while the lack of correlation between common predictors of adoption and uptake is perhaps surprising, the intensiveness of the promotion strategy—with information provision, financing, and subsidies—seems to have allowed less educated and lower wealth households to better understand and finance a new stove purchase.

When we include purchases made during subsequent visits to recover the second and third installments (during which 36 additional households chose to buy stoves, out of the 408 who did not originally buy a stove), we find that the lower purchase rates among class 3 households fade somewhat (columns 4-5). Purchase rates were 16-17% lower than class 1 on average in this analysis (and 8-12% lower than class 2); this is because late adopters were more likely to be in class 3 (and less likely to be in class 2).¹⁸ In considering purchases over a period of multiple visits during which neighboring households were exposed to new stoves, inherent preferences or preference uncertainty may become less important given the influence of peers, increased trust in a promotion intervention, or the potential for learning. The effect of the other controls is unchanged from the first analysis.

The effects of class on the daily use of an intervention stove 3 months after promotion are also somewhat reduced compared to purchase, but class 3 remains 9-12% less likely to be using a stove than class 1 (columns 6-7), and 3-5% less likely than class 2. In addition, perhaps unsurprisingly considering that the DCE was principally designed to shed light on the demand for stove acquisition, the preference classes do not explain use of stoves in the subsample of households who acquired a stove, except for the biomass ICS. Thus, in this setting and with this technology, the influence of preferences appears to work principally through its influence on the purchasing decision (as shown in Appendix Table C5). In interpreting these use results, we note that 76 of the original 1,063 households (7%) were lost from the sample at follow-up (Figure 2). Although attrition is not significantly correlated with treatment status, it may affect the generalizability of our results. Households lost to follow-up were smaller, more likely to be female-headed, and were less reliant on traditional stoves, as measured by time spent cooking, fuel consumption, and time spent collecting fuel at baseline.

¹⁸Specifically, of the 36 households who adopted a stove during these later visits, 29 were in class 3. We estimated probit and OLS regressions to explore the determinants of later purchases, using the same covariates as those in the class determinants analysis; the power of this analysis is limited due to the small number of such households, and we find only weak correlations with greater patience and lower perception of health risk from smoke exposure. Only class 3 membership is consistently and strongly related to lagging purchases.

(1) Basic	(2) + Rebate	(3) + All controls	(4) Basic	(5) + All controls	(6) Basic	(7) + All controls	
Visit 1 purchase	Visit 1 purchase	Visit 1 purchase	With later purchases	With later purchases	Stove use at follow-up [§]	Stove use at follow-up [§]	
0.43^{***}	0.16^{***}	0.16^{***}	0.48^{***}	0.21^{***}	n.a.	n.a.	_ pi
(0.034)	(0.039)	(0.040)	(0.036)	(0.042)			6964
	0.0015^{***}	0.0015^{***}		0.0015^{***}		0.0011^{***}	- 110
	(0.0002)	(0.0002)		(0.0002)		(0.0001)	eΡ
		-0.004		-0.0007		-0.0009	OI
		(0.001) -0.022		(0.001)		(0.001) -0.073	whe
		(0.032)		(0.033)		(0.047)	n cit
		-0.0009		-0.0001		0.0005	tine
		(0.004)		(0.004)		(0.005)	0 *
		0.009		0.008		-0.023	and
		(0.017)		(0.017)		(0.023)	otin
		-0.003		-0.003		0.003	ıσ T
		(0.003)		(0.003)		(0.004)	\mathcal{O}
		-0.030		-0.055*		-0.032	1.1
		(0.030)		(0.028)		(0.036)	htte
0.20^{***}	0.18^{***}	0.19^{***}	0.17^{***}	0.16^{***}	0.090	0.12^{**}	ne./
(0.063)	(0.058)	(0.057)	(0.062)	(0.057)	(0.058)	(0.056)	12
0.14^{**}	0.12^{**}	0.11^{*}	0.12^{**}	0.076	0.045	0.025	ni e
(0.056)	(0.054)	(0.055)	(0.055)	(0.053)	(0.045)	(0.043)	ra/
0.00^{***}	-0.00***	-0.13	0.00^{***}	-0.088	0.27^{***}	-0.021	10
(0.00)	(0.00)	(0.11)	(0.00)	(0.11)	(0.027)	(0.13)	108
No	No	Y_{es}	No	\mathbf{Yes}	No	Yes	3677
987	987	987	987	987	716	716	106C
0.220	0.326	0.341	0.243	0.365	0.005	0.127	37
< 0.05, * p < 0.1. Stair r the full sample with	ndard errors in parent sales intervention dat	heses; standard errors a, however). [†] ·Class 1'	clustered at the and 'Class 2' a:	e hamlet level. The an	alytical sample is denoting assignme	the final sample, aftent to a latent classes	
	(1) Basic Visit 1 purchase 0.43^{***} (0.034) (0.034) (0.034) (0.034) (0.034) (0.034) (0.034) (0.034) (0.034) (0.034) (0.063) (0.0	(1) (2) Basic + Rebate Visit 1 purchase Visit 1 purchase 0.43^{***} 0.16^{***} 0.034) 0.039) 0.034) 0.0015^{***} 0.034) 0.0015^{***} 0.039) 0.002) 0.0015^{***} 0.002) 0.0015^{***} 0.002) 0.0015^{***} 0.002) 0.0015^{***} 0.002) 0.002 0.002 0.002 0.002 0.002 0.002 0.002 0.002 0.002 0.002 0.002 0.002 0.002 0.002 0.002 0.002 0.000^{**} 0.000^{**} 0.000^{**} 0.000^{**} 0.000^{**} 0.000^{**} 0.000^{**} 0.000^{**} 0.000^{**} 0.000^{**} 0.000^{**} 0.000^{**} 0.000^{**} 0.000^{**} 0.000^{**}	(1) (2) (3) Basic + Rebate + All controls Visit 1 purchase Visit 1 purchase Visit 1 purchase 0.43^{***} 0.16^{***} 0.040 0.334 0.039 0.040 0.334 0.0015^{***} 0.002 0.034 0.0015^{***} 0.0015^{***} 0.034 0.0015^{***} 0.0015^{***} 0.031 0.0022 0.0022 0.0015^{***} 0.0012^{***} 0.0012^{***} 0.0023^{**} 0.0012^{***} 0.0012^{***} 0.0023^{***} 0.0012^{****} 0.0023^{***} 0.0033^{****} 0.0033^{***} 0.0033^{***} 0.0033^{***} 0.0033^{***} 0.0033^{***} 0.0033^{***} 0.0033^{***} 0.0033^{**} 0.0003^{***} 0.0033^{***} 0.0033^{**} 0.0033^{***} 0.0033^{**} 0.0033^{**} 0.0033^{**} 0.0033^{***} 0.0033^{**} 0.0003^{**} 0.0033^{**} 0.0033^{**} <td></td> <td></td> <td></td> <td></td>				

the ultimate purchasing behavior should not depend on who participated in the DCE during the baseline survey. None of these controls were found to be significantly related to purchase; as shown they did not alter the sign or significance of the main results (Columns 3, 5 and 7). Observations with missing values for these additional covariates are retained in these regressions by filling in median hamlet-level values; results are not sensitive to this procedure. [§]The use outcome is only analyzed for households who were in the intervention group; hence the smaller sample size.

Table 6: Stove purchase by latent class

This content downloaded from 128.084.124.240 on October 24, 2019 08:30:13 AM All use subject to University of Chicago Press Terms and Conditions (http://www.journals.uchicago.edu/t-and-c).

Copyright The Association of Environmental and Resource Economists 2019. Preprint (not copyedited or formatted).

26

Please use DOI when citing or quoting. DOI: https://doi.org/10.1086/706937 Using two-stage least squares and informed by the results in Table 6, we attempted to predict class membership as a function of the predictors included in those regressions, to test whether predictions of class membership using observables would be informative about ICS purchases. These analyses offer suggestive evidence that households less likely to purchase an ICS (in class 3) could be pre-identified, though standard errors are large. The point estimates indicate that households predicted to be in class 3 are 7-15% less likely to purchase an ICS, consistent with the results presented above (Table 7, columns 1-2). The estimates are unstable, however, and vary greatly with the choice of first stage variables. Similar to the nested F-tests, this tells us that the latent classes convey information beyond easily observed attributes that is highly relevant to household choices.

Question 2: Do different preference types choose different stoves?

Based on differences in the relative weighting of ICS attributes revealed by the DCE, we expected that households in distinct preference classes might choose to purchase alternative technologies at different rates. In particular, class 2 households dislike traditional stoves and have a greater willingness to pay for all stove attributes, while class 1 households place greater weight on smoke emissions relative to other attributes. Moreover, the improved biomass stoves that were offered to households were more expensive; more price sensitive class 1 households might therefore prefer the electric option. The stoves also differed in terms of their fuel costs (the electric stove is likely more costly for many households, unless they place a high value on time savings).

We consider the purchase of different stove types using a multinomial logit model, and then analyze daily use of each stove with an OLS specification (Table 8).¹⁹ Our first observation is that class 1 households are most likely to purchase and use the electric stove (columns 2, 4, and 6); when including all controls, class 1 households are 17% and 25% more likely to purchase this stove than classes 2 and 3, respectively, and 9 and 10% more likely to use it daily than these two classes.²⁰ In contrast, class 2 households appear more likely to purchase and use the biomass-burning ICS (columns 1, 3, and 5); specifically, they are 7% and 5%more likely to purchase a biomass ICS than classes 1 and 3, respectively, and 7 and 8% more likely to use it than classes 1 and 3^{21} As with overall purchase, few other covariates explain differences in purchase and use rates, including those for relative firewood cost and average

¹⁹We again only report results for the first purchase sample. Results from the sample including later purchases are qualitatively identical and are reported in Appendix Table C4.

 $^{^{20}}$ The marginal effects for purchase are evaluated at the mean value of other covariates.

 $^{^{21}}$ Even conditioning on purchase, we find that households in class 3 are also less likely to report daily use of the biomass ICS. In these conditional regressions, we also find that the rebate is positively related to use, which suggests that larger rebates promised to users at the time of the third sales visit also helped to incentivize greater use.

				Ple
Variables	(1) Visit 1 purchase	(2) All purchases	(3) Biomass ICS purchase	(4) Electric ICS purchase of an
Treatment	0.20^{***}	0.25^{***}	0.043	0.20*** • IO
	(0.039)	(0.040)	(0.027))	(0.036) (0.036)
Ireatment × Kebate amount (INK)	(0.002)	(0.0002)	(0.0001)	0.0012 0.002 (0.002)
Predicted Class 1^{\dagger}	0.12	0.15	-0.20*	0.24
	(0.16)	(0.15)	(0.12)	(0.15)
Predicted Class 2^{\dagger}	0.071	0.068	-0.12	0.22* 0
	(0.14)	(0.14)	(0.11)	(0.13)
Constant	-0.032	-0.034	0.052	-0.082**
	(0.045)	(0.041)	(0.034)	(0.041)
Observations	987	987	987	doi.o 286
R^2	0.321	0.343	0.048	0.234
Notes. Two-stage least squares model. *** $_{I}$ The analytical sample is the final sample, aft [†] The controls used to predict class membershi able to save money. Drivate toilet, aware of cl	p < 0.01, ** $p < 0.05$, * $per sample attrition (resulting are: Relative wealth, ag$	< 0.1. Standard errors ts are qualitatively id ge of head of househol at smoke is unsafe fo	ors in parentheses; standard erro- entical for the full sample with s. d, female head of household, gene c health.	ors clustered at the hamlet level ales intervention data, however

Table 7: Two-stage model results for the effect of preference class on purchase

Copyright The Association of Environmental and Resource Economists 2019. Preprint (not copyedited or formatted).

28

Please use DOI when citing or quoting. DOI: https://doi.org/10.1086/706937 hours of electricity supply. Households in the general caste category were somewhat more likely to purchase a biomass ICS, while those who spend more time cooking with traditional stoves and who own private toilets (perhaps reflecting a higher propensity to engage in health risk-averting behavior) were more likely to purchase an electric one.

In additional analysis, we further observed that classes 2 and 3 were somewhat less responsive to the rebate amount for the electric stove only (Appendix Table C4), consistent with the results presented above. And similarly to the previous two-stage analysis, we find some evidence that the households least likely to purchase an electric ICS (class 3) could perhaps be identified using household characteristics, though class 2 households were not so easily identified (Table 7, columns 3-4). As with the analyses of the dichotomous purchase decision, however, these two-stage estimates are unstable and imprecise.

When describing the intervention in Section 3.3, we noted that households might associate different pros and cons with the technologies that were offered to them, and that this might explain why some would favor the biomass ICS while others would opt for the electric stove. At endline, following exposure to these technologies, we surveyed households concerning their opinions of these pros and cons. Beliefs about traditional stoves were mostly unchanged: The main advantages of this technology were the good taste of food (53%), the ability to cook all foods (36%), and its low cost (31%), while key disadvantages were the smoke produced (78%), the amount of fuel required (27%), and the (slow) speed of cooking (22%). Households in the treatment group meanwhile identified clear differences between the electric and biomass ICS. The principal advantages of the former were identified as the (fast) speed of cooking (28%), the attractiveness of the stove (15%), and its portability and low emissions (11% each), while the former was seen as advantageous for its low fuel requirement (25%), portability (19%). and low emissions (17%). Meanwhile, the electric stove's disadvantages were identified as the risk of shocks (34%), the high cost (32%) and unreliability (17%) of electricity, and the stove cost (13%). The biomass ICS, in contrast, had two main disadvantages: high purchase cost (39%), and need for additional fuel processing (i.e., wood chopping) prior to use (11%).

Question 3: What would ICS adoption have been in the absence of stove choice?

The results presented above collectively indicate that higher demand class 2 households preferred biomass-burning ICS' over classes 1 and 3, while smoke- and price-sensitive class 1 households preferred the electric option. These differences across classes suggest that adoption rates might have been lower in the absence of choice. To test this idea, we conducted a set of counterfactual predictions of what would have happened if only one of the new technologies had been offered, using the preference weighting obtained from the DCE analysis (reported in Table 4) and via the following steps. First, we specified the relative levels of the four

Variables	(1) Basic	(2)Basic	(3) + Rebate and Controls	(4) + Rebate and Controls	(5) + Rebate and Controls	(6) + Rebate and Controls
	Biomass ICS purchase	Electric ICS purchase	Biomass ICS purchase	Electric ICS purchase	Biomass ICS use	Electric ICS use
Rebate amount (INR)			0.00066^{***}	0.0011^{***}	0.0004^{***}	0.0005**
~			(0.0001)	(0.0002)	(0.001)	(0.001)
$Class 1^{\dagger}$	-0.042	0.23^{***}	-0.027	0.25^{***}	-0.010	0.087*
	(0.050)	(0.061)	(0.046)	(0.064)	(0.027)	(0.046)
$Class 2^{\dagger}$	0.043	*00.0	0.047	0.075	0.071^{**}	-0.015
	(0.027)	(0.053)	(0.031)	(0.062)	(0.031)	(0.031)
Hours of electricity supply per day			-0.003	-0.0001	0.001	0.004
			(0.004)	(0.006)	(0.002)	(0.003)
Respondent reports high price of firewood	1		0.019	-0.071	0.014	0.004
			(0.024)	(0.042)	(0.021)	(0.030)
Other $controls^{\ddagger}$	No	No	Yes	Yes	Yes	Yes
Observations	716	716	716	716	716	716
$\operatorname{Pseudo-}R^2$	0.017	0.017	0.130	0.130	0.074	0.074
Notes. Columns 1-4 report results from a multinop	mial logit model using initial p	urchase decision only: we repo	ort marginal effects at the mean o	f the sample covariates: columns 5	and 6 are linear probability mode	als for daily use of the biomass and
electric intervention stoves, respectively. *** $p < 0$	0.01, ** p < 0.05, * p < 0.1. St	andard errors in parentheses;	standard errors clustered at the	namlet level. The analytical samp	le is the final sample, after sample	attrition (results are qualitatively
identical for the full sample with sales intervention	data, except that class 2 house	sholds are significantly more li	kely to adopt the biomass ICS wit	h that larger sample). [†] Class 1 an	d Class 2 are indicator variables de	moting assignment to latent classe
1 and 2, respectively. Class 3 is omitted. $^{\ddagger}The oth$	ner controls include all of those	from the complete model in	Table 6 (e.g., the basic controls fi	om Table 6, column 3 plus those i	indicated in the notes below Table	6). Few of these were found to b
significantly related to purchase; as shown they did	I not alter the sign or significar	nce of the main results shown	here (columns 3-6). Observations	with missing values for these addi	tional covariates are retained in th	ese regressions by filling in mediat
values; results are not sensitive to this procedure.						
						100

Copyright The Association of Environmental and Resource Economists 2019. Preprint (not copyedited or formatted). Please use DOI when citing or quoting. DOI: https://doi.org/10.1086/706937

30

Please use DOI when citing or quoting. DOI: https://doi.org/10.1086/706937 attributes corresponding to our two ICS alternatives, also incorporating the price variation arising from the three rebate levels. Second, we assumed that if only one of the two stoves had been offered, those already purchasing each of the ICS in our real-world experiment would still have selected it. In other words, removing the alternative not favored by specific households would not have affected their purchasing decision, given that they had been observed to favor that remaining option anyway. Similarly, those who had opted out of stove purchases entirely would not have selected the remaining option that they did not favor when given the choice of two ICS. Thus, this left a group of households whose preferred ICS was assumed to no longer be available in our counterfactual. For this group of households, we derived class-specific probabilities of adoption for the remaining ICS, based on the relative probabilities obtained from that class' preference weighting for it versus the opt-out alternative. We conducted these calculations for the low and high prices in our field experiment (that is, the price of the each ICS was allowed to vary between the lowest and highest rebate amounts, affecting the probability of adoption, as shown in Table 9). We then assigned this sub-group of households from each class to the two remaining alternatives - the single ICS and the opt-out status quo - in direct proportion to these derived relative probabilities.²²

We present these probabilities as a percentage of all households in a given preference class in Table 9, for the two counterfactual situations: electric ICS only, and biomass ICS only. The predictions of the stove purchases that would have been made in the situation where only one or the other ICS would have been presented following re-assignments based on these choice probabilities then appears in Figure 4.²³ These calculations suggest that purchase rates would have been 5-14 percentage points lower if only the electric ICS had been available, and 13-34 percentage points lower with only the biomass ICS. The high end of this range corresponds to a higher rebate, since purchase rates are increasing in this price discount.

5 Discussion

In this study, we considered the hidden nature and latent preferences for new technology, and then assessed whether this source of heterogeneity was related to actual purchasing decisions. To the best of our knowledge, it is the first study mapping preferences for technology to

 $^{^{22}}$ An alternative setup for this calculation would use the hypothetical DCE data to make predictions under two hypothetical scenarios - one with and one without stove choice. In this approach, we would be putting a great deal of weight on the attributes that were included in the DCE, and especially on the alternative-specific constant for a biomass ICS. We feel that this is less compelling than use of both the revealed preference information and the weights from the DCE.

 $^{^{23}}$ Clearly then, this calculation assumes that stove selection preferences would not have been affected by the sheer presence of choice, which might occur if households selected a particular stove solely because they could compare it to the other option being offered.

Figure 4: Purchase of ICS stoves, revealed in the experiment, and for "no choice" counterfactual



Table 9: Choice probabilities for preference class-ICS combinations, at high and low price levels

Electric stove only	Low ICS Price	(INR 640)	High ICS	price (INR 935)
	Electric ICS	Traditional	Electric ICS	Traditional
Class 1 Class 2 Class 3	$rac{86\%}{71\%}$ 21%	14% 29% 79%	$68\% \\ 70\% \\ 17\%$	$32\% \\ 30\% \\ 83\%$
Biomass stove only	Low ICS Price	(INR 720)	High ICS pr	tice (INR 1,055)
	Biomass ICS	Traditional	Biomass ICS	Traditional
Class 1	44% 57%	56%	19%	81%
Class 2		43%	56%	44%

Notes. Probabilities are for purchase of the stove given only the alternative traditional stove. For smoke, we assume zero smoke emissions for the electric ICS, a 20% reduction for the biomass ICS. We further assume the biomass ICS reduces fuel requirements by 25%, whereas the electric ICS has maximum fuel requirements.

Please use DOI when citing or quoting. DOI: https://doi.org/10.1086/706937 experimentally-induced choices. As such, our work draws on the literature that compares or combines data from stated and revealed preference measures of demand (Adamowicz et al., 1994; Carson et al., 1996; Hensher and Bradley, 1993), but extends it in a new and different way. Specifically, we test the hypothesis that attitudes characterized using stated preference methods can predict technology adoption.

Our specific application considers the challenge of clean energy technology adoption in Uttarakhand, India. The use of traditional, inefficient and highly polluting cooking technologies is ubiquitous in this setting, but more generally affects billions of the world's poorest people (Legros et al., 2009; Smith et al., 2000). These technologies cause illness (Bruce et al., 2006; Ezzati and Kammen, 2001; Martin et al., 2011), unsustainable and timeintensive harvesting of biomass, and environmental damage (Bond et al., 2004; Ramanathan and Carmichael, 2008). Many have proposed that solutions available to households are not sufficiently adapted to local cooking requirements and user preferences (Bishop et al., 2013; Cordes, 2011; Duflo et al., 2008; Jeuland and Pattanayak, 2012; Lewis and Pattanayak, 2012; Singh and Pathy, 2012). In India, for example, penetration and impacts of improved cooking technology have been limited despite decades of subsidized promotion and the largest potential market for cleaner cooking technologies in the world (Hanna et al., 2016).

In this setting, we found that household preferences for non-traditional stoves were highly varied. Latent class analysis identified three broad classes of respondents: two classes, comprising 12% and 23% of the sample, who appeared to be differentially 'interested' in the features of cleaner technology, whereas a large third class (65% of households) was generally 'uninterested' in these attributes. Within the first two classes, class 1 placed greater relative weight on smoke emissions reductions, whereas class 2 was less price sensitive and equally valued emission reductions, fuel savings and greater convenience. Closer examination of the make-up of each class showed that the 'uninterested' class mainly consisted of lower-SES households who were less aware of clean options and the harmful effects of smoke inhalation.

Turning to the relationship between preferences and adoption decisions during a randomized stove promotion intervention, we found that households in the 'uninterested' class were less likely to purchase and use a new stove, despite the fact that the promotion effort intensively provided information and conducted in-house demonstrations. These recalcitrant class 3 households were also least responsive to incentives. The silver lining in these results, however, is that class membership became somewhat less predictive over time, as the initially 'uninterested' households were more likely to adopt a new stove during later sales visits. And while our study was not designed to allow determination of the precise mechanism behind this changing of minds, this result provides hope that these households' decisions can evolve.

We further noted that class 1 households—who placed the greatest relative weight on

Please use DOI when citing or quoting. DOI: https://doi.org/10.1086/706937 reduced smoke emissions, and had lower WTP for the improved biomass stoves offered in

the DCE—were most likely to adopt an electric ICS, which is both cheaper and offers the possibility of eliminating cooking smoke. Critically, the predictive power of preference class was retained even when controlling for a large set of household baseline covariates (e.g., expenditures, perceived relative wealth, education, age) that are typically associated with adoption. Also importantly, using those covariates to predict class membership yielded point estimates of the effects of predicted class membership on adoption that were not inconsistent with the experimental results, though the estimates were imprecise. Thus, interventions may be able to improve targeting by working to characterize preferences ex ante. Or they may offer alternatives to beneficiaries instead: counterfactual calculations indicate that adoption would have been depressed in the absence of choice.

Our findings thus offer new insights that could improve efforts to promote other beneficial consumer products and quasi-public goods. An oft-neglected feature of the challenge of technology provision in low-income settings is the fact that governments, NGOs or other practitioners must frequently act as the primary suppliers, which typically results in limited (or no) options for consumers. Private markets meet heterogeneous demand through market segmentation and product differentiation, but the competitive forces that drive such innovations are typically lacking when markets—such as those for quasi-public goods, and especially in developing countries—are thin. In addition, implementers and practitioners (whether NGOs or governments) are often tempted to pick specific solutions based on technical criteria (e.g., an emissions profile). These well-meaning implementers may have aims that are not wholly compatible with beneficiaries' desires - for example, the cookstove sector is dominated by rather dogmatic views about appropriate technology, based on strong beliefs about which fuels or which stoves are able to deliver health benefits. Such a top-down supply-driven model has failed over and over in multiple sectors (Pritchett and Woolcock, 2004). In this sense, support for market-based provision of innovations that meet consumer preferences while providing positive spillovers, such as results-based financing, are a potentially promising approach.

It is also expensive for implementers to offer choices, since an entire supply chain—in our case, manufacturers, wholesalers, retailers, financiers, maintenance, and fuel suppliers—must often be repurposed to do so. Such costs must be weighed against the value of choice, which will vary considerably across contexts and technologies. Moreover, political feasibility often dictates delivery of simple interventions. It can also be difficult to craft programs with messages that do not confuse target beneficiaries, especially when programs entail complex tradeoffs and/or lead to unintended negative outcomes. For example, an option that is offered may be popular but ineffective, or an intervention may be designed around a technology that

Please use DOI when citing or quoting. DOI: https://doi.org/10.1086/706937 only provides a specific type of benefit (e.g., health, or climate mitigation) at the expense of others. Still, even in contexts where choices appear to not be preferred for political or cost reasons, we must also assess the the downside of limiting choices. Indeed, studies to uncover preference heterogeneity could be used to help policymakers understand how uptake may be affected by constraining options, and what, if any options warrant additional investment and marketing. By clearly demonstrating that latent preferences influence technology adoption, our research points to the need for more seriously considering behaviors and choices rather than prescribing technical fixes that may not align with household needs.

References

- Adamowicz, W., J. Louviere, and M. Williams, "Combining Revealed and Stated Preference Methods for Valuing Environmental Amenities," Journal of Environmental *Economics and Management*, may 1994, *26* (3), 271–292. (Not cited.)
- Allcott, H. and S. Mullainathan, "Behavior and Energy Policy," Science, mar 2010, 327 (5970), 1204–1205. (Not cited.)
- Ashraf, N., D. Karlan, and W. Yin, "Tying Odysseus to the Mast: Evidence From a Commitment Savings Product in the Philippines," The Quarterly Journal of Economics, may 2006, 121 (2), 635–672. (Not cited.)
- Bensch, Gunther, Michael Grimm, and Jörg Peters, "Why do households forego high returns from technology adoption? Evidence from improved cooking stoves in Burkina Faso," Journal of Economic Behavior & Organization, aug 2015, 116, 187–205. (Not cited.)
- Bertrand, Marianne and Sendhil Mullainathan, "Do People Mean What They Say? Implications for Subjective Survey Data," American Economic Review, may 2001, 91 (2), 67–72. (Not cited.)
- Bhojvaid, Vasundhara, Marc Jeuland, Abhishek Kar, Jessica Lewis, Subhrendu Pattanayak, Nithya Ramanathan, Veerabhadran Ramanathan, and Ibrahim **Rehman**, "How do People in Rural India Perceive Improved Stoves and Clean Fuel? Evidence from Uttar Pradesh and Uttarakhand," International Journal of Environmental *Research and Public Health*, jan 2014, *11* (2), 1341–1358. (Not cited.)
- Bishop, Simon, Pradeep Pursnani, and Colin Sumpter, "Social Marketing in India: Lessons learned from efforts to foster demand for cleaner cookstoves," Technical Report, Shell Foundation, London 2013. (Not cited.)
- Bonan, J., P. LeMay-Boucher, and M. Tenikue, "Increasing anti-malaria bednet takeup using information and distribution strategies: evidence from a field trial in Senegal," Journal of Development Effectiveness, 2017, 9 (4), 543–562. (Not cited.)
- Bond, Tami, Chandra Venkataraman, and Omar Masera, "Global atmospheric impacts of residential fuels," Energy for Sustainable Development, sep 2004, 8 (3), 20–32. (Not cited.)

- Borghans, Lex, Angela Lee Duckworth, James J. Heckman, and Bas ter Weel, "The Economics and Psychology of Personality Traits," Journal of Human Resources, 2008, 43 (4), 972–1059. (Not cited.)
- Brown, Joe, Amar Hamoudi, Marc Jeuland, and Gina Turrini, "Seeing, believing, and behaving: Heterogeneous effects of an information intervention on household water treatment," Journal of Environmental Economics and Management, nov 2017, 86, 141–159. (Not cited.)
- Bruce, Nigel, Eva Rehfuess, Sumi Mehta, Guy Hutton, and Kirk Smith., "Indoor Air Pollution," in Dean T Jamison, Joel G Breman, Anthony R Measham, George Alleyne, Mariam Claeson, David B Evans, Prabhat Jha, Anne Mills, and Philip Musgrove, eds., Disease Control Priorities in Developing Countries, 2 ed., Washington, DC: World Bank, 2006, chapter 42, pp. 793–815. (Not cited.)
- Cárdenas, Juan Camilo, Alberto Chong, and Hugo Nopo, "Stated social behavior and revealed actions: Evidence from six Latin American countries," Journal of Development *Economics*, sep 2013, 104, 16–33. (Not cited.)
- Carson, Richard T., Nicholas E. Flores, Kerry M. Martin, and Jennifer L. Wright, "Contingent Valuation and Revealed Preference Methodologies: Comparing the Estimates for Quasi-Public Goods," Land Economics, feb 1996, 72 (1), 80. (Not cited.)
- Casabonne, Ursula and Charles Kenny, "The Best Things in Life are (Nearly) Free: Technology, Knowledge, and Global Health," World Development, jan 2012, 40 (1), 21–35. (Not cited.)
- **Cordes, Leslie**, "Igniting change: A strategy for universal adoption of clean cookstoves and fuels," Technical Report, Global Alliance for Clean Cookstsoves, Washington, DC 2011. (Not cited.)
- Duflo, Esther, Michael Greenstone, and Rema Hanna, "Indoor air pollution, health and economic well-being," Surveys and Perspectives Integrating Environment and Society (S.A.P.I.E.N.S), 2008, 1 (1). (Not cited.)
- Ezzati, Majid and Daniel M Kammen, "Indoor air pollution from biomass combustion and acute respiratory infections in Kenya: an exposure-response study," The Lancet, aug 2001, 358 (9282), 619–624. (Not cited.)
- Feder, Gershon, Richard E. Just, and David Zilberman, "Adoption of Agricultural Innovations in Developing Countries: A Survey," Economic Development and Cultural *Change*, jan 1985, *33* (2), 255–298. (Not cited.)
- Foster, Andrew D. and Mark R. Rosenzweig, "Microeconomics of Technology Adoption," Annual Review of Economics, sep 2010, $\mathcal{Z}(1)$, 395–424. (Not cited.)
- Hanna, Rema, Esther Duflo, and Michael Greenstone, "Up in Smoke: The Influence of Household Behavior on the Long-Run Impact of Improved Cooking Stoves," American Economic Journal: Economic Policy, feb 2016, 8 (1), 80–114. (Not cited.)
- Heckman, James J., "Micro Data, Heterogeneity, and the Evaluation of Public Policy: Nobel Lecture," Journal of Political Economy, aug 2001, 109 (4), 673–748. (Not cited.)
- Hensher, David A. and Mark Bradley, "Using stated response choice data to enrich revealed preference discrete choice models," Marketing Letters, apr 1993, 4 (2), 139–151.

(Not cited.)

- Hollada, J., K. Williams, C. Miele, D. Danz, S. Harvey, and W. Checkley, "Perceptions of improved biomass and liquefied petroleum gas stoves in Puno, Peru: implications for promoting sustained and exclusive adoption of clean cooking technologies," *International Journal of Environmental Research and Public Health*, 2017, 14 (2), 182. (Not cited.)
- Jenkins, Marion W. and Val Curtis, "Achieving the 'good life': Why some people want latrines in rural Benin," *Social Science & Medicine*, dec 2005, 61 (11), 2446–2459. (Not cited.)
- Jeuland, M., J. Orgill, A. Shaheed, G. Revell, and J. Brown, "A matter of good taste: investigating preferences for in-house water treatment in peri-urban communities in Cambodia," *Environment and Development Economics*, 2016, 21 (3), 291–317. (Not cited.)
- Jeuland, M.A., V. Bhojvaid, A. Kar, J.J. Lewis, O. Patange, S.K. Pattanayak, N. Ramanathan, I.H. Rehman, J.S. Tan Soo, and V. Ramanathan, "Preferences for improved cook stoves: Evidence from rural villages in north India," *Energy Economics*, nov 2015, 52, 287–298. (Not cited.)
- Jeuland, Marc A. and Subhrendu K. Pattanayak, "Benefits and Costs of Improved Cookstoves: Assessing the Implications of Variability in Health, Forest and Climate Impacts," *PLoS ONE*, feb 2012, 7 (2), e30338. (Not cited.)
- Jeuland, Marc, Subhrendu K. Pattanayak, and Randall Bluffstone, "The Economics of Household Air Pollution," Annual Review of Resource Economics, oct 2015, 7(1), 81–108. (Not cited.)
- Krishnapriya, P.P. and R. Somanathan, "Effects of information on energy choices in rural India," 2019. Working paper. (Not cited.)
- Lambrecht, I., M. Maertens, L. Vranken, R. Merckx, and B. Vanlauwe, "Heterogeneous preferences for integrated soil fertility management: a choice experiment on climbing beans in Burundi," 2013. University of Leuven Bioeconomics Working Paper Series. (Not cited.)
- Larson, H.J., C. Jarrett, E. Eckersberger, D.M. Smith, and P. Paterson, "Understanding vaccine hesitancy around vaccines and vaccination from a global perspective: a systematic review of published literature," *Vaccine*, 2014, *32* (19), 2150–2159. (Not cited.)
- Legros, Gwénaëlle, Ines Havet, Nigel Bruce, and Sophie Bonjour, "The Energy Access Situation in Developing Countries: A Review Focusing on the Least Developed Countries and Sub-Saharan Africa," Technical Report, UNDP and World Health Organization, New York 2009. (Not cited.)
- Lewis, Jessica J. and Subhrendu K. Pattanayak, "Who Adopts Improved Fuels and Cookstoves? A Systematic Review," *Environmental Health Perspectives*, feb 2012, *120* (5), 637–645. (Not cited.)
- -, Vasundhara Bhojvaid, Nina Brooks, Ipsita Das, Marc A. Jeuland, Omkar Patange, and Subhrendu K. Pattanayak, "Piloting Improved Cookstoves in India," *Journal of Health Communication*, mar 2015, 20 (sup1), 28–42. (Not cited.)
- Liu, Elaine M., "Time to Change What to Sow: Risk Preferences and Technology Adoption Decisions of Cotton Farmers in China," *Review of Economics and Statistics*, oct 2013, 95

- Louviere, Jordan J., David A. Hensher, and Joffre D. Swait, *Stated Choice Methods: Analysis and Applications*, Cambridge USA: Cambridge University Press, 2000. (Not cited.)
- Madrian, B. C. and D. F. Shea, "The Power of Suggestion: Inertia in 401(k) Participation and Savings Behavior," *The Quarterly Journal of Economics*, nov 2001, *116* (4), 1149–1187. (Not cited.)
- Magidson, Jay and Jeroen K. Vermunt, "Latent Class Models," in David Kaplan, ed., The SAGE Handbook of Quantitative Methodology for the Social Sciences, Thousand Oaks: SAGE Publications, Inc, 2004, chapter 10, pp. 175–198. (Not cited.)
- and _ , "Removing the Scale Factor Confound in Multinomial Logit Choice Models to Obtain Better Estimates of Preference," in "Proceedings of the Sawtooth Software Conference" Sawtooth Software 2007, pp. 139–174. (Not cited.)
- Manski, Charles F., "Measuring Expectations," *Econometrica*, sep 2004, 72 (5), 1329–1376. (Not cited.)
- Martin, W. J., R. I. Glass, J. M. Balbus, and F. S. Collins, "A Major Environmental Cause of Death," *Science*, oct 2011, *334* (6053), 180–181. (Not cited.)
- McFadden, Daniel and Kenneth Train, "Mixed MNL models for discrete response," Journal of Applied Econometrics, 2000, 15 (5), 447–470. (Not cited.)
- McFadden, Daniel L., "Econometric Models of Probabilistic Choice," in Charles F. Manski and Daniel L. McFadden, eds., *Structural Analysis of Discrete Data with Econometric Applications*, Cambridge USA: The MIT Press, 1981, pp. 198–272. (Not cited.)
- Miller, G. and A.M. Mobarak, "Gender differences in preferences, intra-household externalities, and low demand for improved cookstoves," 2013. National Bureau of Economic Research Working Paper No. w18964. (Not cited.)
- Mulatu, D.W., A. van der Veen, and P.R. van Oel, "Farm households' preferences for collective and individual actions to improve water-related ecosystem services: The Lake Naivasha basin, Kenya," *Ecosystem Services*, 2014, 7, 22–33. (Not cited.)
- Murphy, James J., P. Geoffrey Allen, Thomas H. Stevens, and Darryl Weatherhead, "A Meta-analysis of Hypothetical Bias in Stated Preference Valuation," *Environmental & Resource Economics*, mar 2005, 30 (3), 313–325. (Not cited.)
- Orgill, Jennifer, Ameer Shaheed, Joe Brown, and Marc Jeuland, "Water quality perceptions and willingness to pay for clean water in peri-urban Cambodian communities," *Journal of Water and Health*, sep 2013, 11 (3), 489. (Not cited.)
- Pattanayak, S.K., M. Jeuland, J.J. Lewis, F. Usmani, N. Brooks, V. Bhojvaid, A. Kar, L. Lipinski, L. Morrison, O. Patange, N. Ramanathan, I.H. Rehman, R. Thadani, M. Vora, and V. Ramanathan, "Experimental evidence on promotion of electric and improved biomass cookstoves," *Proceedings of the National Academy of Sciences*, 2019, 116 (27), 13282–13287. (Not cited.)
- Phaneuf, Daniel J., "Heterogeneity in Environmental Demand," Annual Review of Resource Economics, jun 2013, 5 (1), 227–244. (Not cited.)
- Pinto, Rodrigo, "Selection Bias in a Controlled Experiment: The Case of Moving to

Opportunity.," 2015. Mimeo. (Not cited.)

- Poe, Gregory L. and Richard C. Bishop, "Valuing the Incremental Benefits of Groundwater Protection when Exposure Levels are Known," *Environmental and Resource Economics*, 1999, 13 (3), 341–367. (Not cited.)
- **Pritchett, Lant and Michael Woolcock**, "Solutions When *the* Solution is the Problem: Arraying the Disarray in Development," *World Development*, feb 2004, *32* (2), 191–212. (Not cited.)
- Ramanathan, V. and G. Carmichael, "Global and regional climate changes due to black carbon," *Nature Geoscience*, mar 2008, 1 (4), 221–227. (Not cited.)
- Ravallion, Martin, "Poor, or Just Feeling Poor? On Using Subjective Data in Measuring Poverty," 2012. World Bank Policy Research Working Paper № 5968. (Not cited.)
- Revelt, David and Kenneth Train, "Mixed Logit with Repeated Choices: Households' Choices of Appliance Efficiency Level," *Review of Economics and Statistics*, nov 1998, 80 (4), 647–657. (Not cited.)
- Roeder, Kathryn, Kevin G. Lynch, and Daniel S. Nagin, "Modeling Uncertainty in Latent Class Membership: A Case Study in Criminology," *Journal of the American Statistical Association*, sep 1999, 94 (447), 766–776. (Not cited.)
- Singh, Arunesh and Suma Pathy, "Promoting advanced cook stoves through a sustainable partnership between a micro finance institution and a syndicated distribution network: The advanced cook stoves initiative," Technical Report, Abt Associates: Market-based partnerships for health, New Delhi 2012. (Not cited.)
- Smith, Kirk R., R. Uma, V.V.N. Kishore, Junfeng Zhang, V. Joshi, and M.A.K. Khalil, "Greenhouse Implications of Household Stoves: An Analysis for India," Annual Review of Energy and the Environment, nov 2000, 25 (1), 741–763. (Not cited.)
- Smith, V. Kerry and William H. Desvousges, "Averting behavior: Does it exist?," *Economics Letters*, jan 1986, 20 (3), 291–296. (Not cited.)
- Swait, Joffre and Jordan Louviere, "The Role of the Scale Parameter in the Estimation and Comparison of Multinomial Logit Models," *Journal of Marketing Research*, aug 1993, 30 (3), 305. (Not cited.)
- Tanaka, Tomomi, Colin F. Camerer, and Quang Nguyen, "Risk and Time Preferences: Linking Experimental and Household Survey Data from Vietnam," in "Behavioral Economics of Preferences, Choices, and Happiness," Springer Japan, 2016, pp. 3–25. (Not cited.)
- Thiene, Mara, Riccardo Scarpa, and Jordan J. Louviere, "Addressing Preference Heterogeneity, Multiple Scales and Attribute Attendance with a Correlated Finite Mixing Model of Tap Water Choice," *Environmental and Resource Economics*, nov 2014, 62 (3), 637–656. (Not cited.)
- UN Foundation, "Secretary Clinton Announces Global Alliance for Clean Cookstoves," 2010. http://www.unfoundation.org/news-and-media/press-releases/2010/ secretary-clinton-announces-global-alliance-for-clean-cookstoves.html. (Not cited.)

Vasquez, W.F. and J. Alicea-Planas, "Unbundling household preferences for improved

Please use DOI when citing or quoting. DOI: https://doi.org/10.1086/706937 sanitation: A choice experiment from an urban settlement in Nicaragua," Journal of Environmental Management, 2018, 218, 477–485. (Not cited.)

- Whittington, Dale, Marc Jeuland, Kate Barker, and Yvonne Yuen, "Setting Priorities, Targeting Subsidies among Water, Sanitation, and Preventive Health Interventions in Developing Countries," *World Development*, aug 2012, 40 (8), 1546–1568. (Not cited.)
- Yang, Jui-Chen, Subhrendu K. Pattanayak, F. Reed Jonson, Carol Mansfield, Caroline van den Berg, and Kelly Jones, "Unpackaging Demand For Water Service Quality: Evidence From Conjoint Surveys In Sri Lanka," 2006. World Bank Policy Research Working Paper № 3817. (Not cited.)